# Druggable Protein Interaction Sites Are More Predisposed to Surface Pocket Formation than the Rest of the Protein Surface

# David K. Johnson<sup>1</sup>, John Karanicolas<sup>1,2</sup>\*

1 Center for Bioinformatics, University of Kansas, Lawrence, Kansas, United States of America, 2 Department of Molecular Biosciences, University of Kansas, Lawrence, Kansas, United States of America

# Abstract

Despite intense interest and considerable effort via high-throughput screening, there are few examples of small molecules that directly inhibit protein-protein interactions. This suggests that many protein interaction surfaces may not be intrinsically "druggable" by small molecules, and elevates in importance the few successful examples as model systems for improving our fundamental understanding of druggability. Here we describe an approach for exploring protein fluctuations enriched in conformations containing surface pockets suitable for small molecule binding. Starting from a set of seven unbound protein structures, we find that the presence of low-energy pocket-containing conformations is indeed a signature of druggable protein interaction sites and that analogous surface pockets are not formed elsewhere on the protein. We further find that ensembles of conformations generated with this biased approach structurally resemble known inhibitor-bound structures more closely than equivalent ensembles of unbiased conformations. Collectively these results suggest that "druggability" is a property encoded on a protein surface through its propensity to form pockets, and inspire a model in which the crude features of the predisposed pocket(s) restrict the range of complementary ligands; additional smaller conformational changes then respond to details of a particular ligand. We anticipate that the insights described here will prove useful in selecting protein targets for therapeutic intervention.

Citation: Johnson DK, Karanicolas J (2013) Druggable Protein Interaction Sites Are More Predisposed to Surface Pocket Formation than the Rest of the Protein Surface. PLoS Comput Biol 9(3): e1002951. doi:10.1371/journal.pcbi.1002951

Editor: James M. Briggs, University of Houston, United States of America

Received August 7, 2012; Accepted January 11, 2013; Published March 7, 2013

**Copyright:** © 2013 Johnson, Karanicolas. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Funding:** This work was supported by grants from the National Center for Research Resources (5P30RR030926) and the National Institute of General Medical Sciences (1R01GM099959 and 8P30GM103495) of the National Institutes of Health, and the Alfred P. Sloan Fellowship (JK). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

\* E-mail: johnk@ku.edu

# Introduction

Manipulating the interactions between proteins represents a promising avenue for therapeutic intervention in a variety of settings. Given the ubiquitous nature of protein interactions, selectively manipulating such interactions could serve as a means to treat conditions including viral and bacterial infections, cancer, and autoimmune disorders [1–7]. In spite of recent ongoing efforts that have provided cause for optimism, protein interactions continue to be viewed as a challenging class of therapeutic target [8–12]. While high-throughput screening efforts that fail to yield extensive hits are typically not reported in the literature, hit rates as low as 0.01% in a large pharmaceutical library have been described [13].

This dearth of successful representatives to study has given increased importance to the several cases in which a protein structure has been solved in complex with a biological protein partner and also in complex with a small molecule inhibitor. Wells and McClendon [8] compared six such cases and observed that binding was not associated with a large conformational change in any of these examples; and yet, the concave pocket on the protein surface at which the small molecule binds was typically smaller or not present in the unbound protein structure. In order for inhibitor binding to occur, the surface of the unbound structure therefore had to undergo local rearrangement to reveal a small molecule binding site that would not necessarily be evident from the unbound structure [8].

Given the limited success in identifying modulators of proteinprotein interactions, it has proven helpful at an early stage to validate a protein surface site by evaluating its "druggability". As such, fragment-based methods have been developed to experimentally assess the druggability of a protein interaction site by determining which members of a small molecule probe set bind to a target protein, and where on the protein surface these bind. This experiment can be conducted using "SAR by NMR" [14], which tracks chemical shift differences to identify binding sites on the protein surface, or by the "multiple solvent crystal structures" method [15], in which independent structures of the target protein are solved after soaking with a collection of organic solvents. Both of these approaches aim to probe the regions of a protein surface that can accommodate small-molecule binding, with a preference for sites that are not uniquely disposed to bind a particular preselected ligand. In either technique, specific regions of the protein surface that interact with a variety of probe molecules - albeit weakly - are inferred to be a putative site for more potent binding by some yet unidentified compound.

Inspired by these methods, we hypothesize that the ability to form a binding pocket may be the limiting factor for druggability

# **Author Summary**

Identifying small-molecule inhibitors of protein interactions has traditionally presented a challenge for modern screening methods, despite interest stemming from the fact that such interactions comprise the underlying mechanisms for cell proliferation, differentiation, and survival. This suggests that many protein interaction surfaces may not be intrinsically "druggable" by small molecules, and elevates in importance the few successful examples as model systems for improving our understanding of factors contributing to druggability. Here we describe a new approach for exploring protein fluctuations leading to surface pockets suitable for small molecule binding. We find that the presence of such pockets is indeed a signature of druggable protein interaction sites, suggesting that "druggability" is a property encoded on a protein surface through its propensity to form pockets. We anticipate that the insights described here will prove useful in selecting protein targets for therapeutic intervention.

of a protein surface site. We further propose that compounds identified in biochemical screens as inhibitors of protein interactions result from natural shape complementarity to specific surface pockets that form with little energetic cost to the protein. Together these hypotheses imply that druggable sites differ from the rest of the protein surface, in that fluctuations under physiological conditions at druggable sites include a special subset of "pocketcontaining" conformations.

To test these hypotheses, we have developed computational methodology to explore protein fluctuations in a biased way, by providing a driving force towards conformations in which a surface pocket is present. Several other studies have generated ensembles of protein conformations reflecting fluctuations around the native state and used these either to assess druggability [16] or as a starting point for docking studies [17-24]. Because these ensembles are generated in an unbiased manner, however, a large fraction of the resulting ensembles correspond to protein conformations that are dissimilar to the corresponding bound structure in both pocket size and hydrophobicity [18]. One of these studies found that carrying out simulations in methanol led to formation of surface pockets which could accommodate small molecules [25], but the use of this non-biological solvent may lead to unphysical artifacts in the resulting models. Molecular dynamics has also been used in a computational analog of "SAR by NMR" in which simulations are carried out using an explicit mixed solvent, allowing druggable sites to be identified by locating accumulation of probe molecules [17,20,26]. Though these methods proved effective for binding sites, the unbiased nature of the underlying simulations make these approaches very computationally intensive. Recently Kozakov et al. have developed a computational analog of the multiple solvent crystal structures method, by using docking to identify "consensus" sites at which several probe molecules cluster [27]. They confirmed that these molecular probes indeed cluster at established druggable sites and that known inhibitors often occupy these consensus sites. Such an approach, however, cannot efficiently explore surface pockets that form via concerted motions involving the protein backbone due to the computational expense associated with repeatedly docking multiple small molecule probes. The biasing potential we describe here avoids this limitation by not needing to dock probe molecules, and therefore can be used in the course of a simulation that samples a broader range of conformational fluctuations.

# Results

#### Quantitative analysis of surface pockets

Because a wide, shallow pocket that is unsuitable for small molecule binding can have the same volume as a deep pocket that is more suitable for small molecule binding, we introduce the concept of "deep" volume of a pocket: the volume of the pocket that is well-sequestered away from solvent. To quantitatively identify small molecule binding pockets and measure their "deep' volume, we implemented a modified version of the LIGSITE<sup>cs</sup> algorithm [28]. This approach starts by creating a grid around a protein and marking each grid point as occupied by protein, surface, or solvent. Next, the algorithm performs linear searches on the grid to find "Surface-Solvent-Surface" events: lines drawn between two surface points that pass through only solvent (Figure **S1**), indicating a concave region on the protein surface. To distinguish between total pocket volume and deep volume, pocket points that fall within 2.5 Å of solvent are marked as "surface pocket" points and are excluded from the "deep volume" calculation. Finally, the remaining contiguous points involved in these events are clustered into "deep pockets". As expected, the deep pocket volumes we use here are correlated to, but smaller than, pocket volumes found by other pocket detection methods, such as Q-SiteFinder [19] (Figure S2). Our implementation differs from the original LIGSITE<sup>cs</sup> algorithm [28] in that our search is restricted to the region around a specific "target" residue on the protein surface, allowing us to rapidly test for pockets at a specific surface site. Additional minor differences are described in Text S1.

A demonstration of this method is shown in **Figure 1A**. Bcl-X<sub>L</sub> is an anti-apoptotic protein in the Bcl-2 family whose overexpression has been implicated in the survival of cancer cells. A series of acyl-sulfonamide-based ligands have been shown to inhibit Bcl-X<sub>L</sub> activity by competing for its peptide-binding groove. Here, we have removed one such inhibitor from a cocrystal structure and applied our modified implementation of the LIGSITE<sup>cs</sup> algorithm at this surface site. The resulting pocket has intuitive shape complementarity to the ligand even though it was generated from the protein structure without the ligand present. This is unsurprising, given that the ligand occupying this pocket is complementary in shape to the protein surface.

We compiled a test set of all seven proteins for which structures have been solved both alone and in complex with a small-molecule inhibitor bound to a protein interaction site (Bcl-X<sub>L</sub>, IL-2, FKBP12, HPV E2, ZipA, MDM2, and the BIR3 domain of



**Figure 1. Identifying surface binding pockets.** (A)  $Bcl-X_L$  (*grey surface*) is shown in complex with an inhibitor (*blue sticks*). The protein surface features a large pocket (*red spheres*) complementary in shape to the inhibitor. (B) Deep pocket volumes of surface pockets at protein interaction sites harboring a bound inhibitor (*red line*) are larger than those found elsewhere on the protein surface (*black line*). Data are collected from a test set of seven proteins, each of which has been solved in complex with a small-molecule inhibitor (Bcl-X<sub>L</sub>, IL-2, FKBP12, HPV E2, ZipA, MDM2, and the BIR3 domain of XIAP). doi:10.1371/journal.pcbi.1002951.g001

XIAP) (**Table 1**). We compared the deep pocket volume at randomly selected regions of the protein surface (see **Text S1**) to the deep volume of the inhibitor-bound pocket. Since the pocket definition can depend somewhat on the particular "target" residue used, the deep volume of the inhibitor-bound pocket was measured several times using different target residues. As shown in **Figure 1B**, across all seven proteins, most pockets identified by this algorithm at randomly selected sites (*black line*) on the protein surface have a deep volume smaller than 25 Å<sup>3</sup>, and all are smaller than 75 Å<sup>3</sup>. In contrast, the distribution of inhibitor-bound pocket volumes (*red line*) is significantly shifted, with about half of the inhibitor-bound deep pocket volumes larger than 75 Å<sup>3</sup>. This observation is consistent with results generated using other pocket detection methods [29–33], although those other studies were not focused on inhibitors of protein-protein interactions.

#### Druggability of protein surface sites

The results presented above demonstrate that inhibitor binding occurs at surface sites containing a pocket, and that these sites are distinct from the remainder of the protein surface. We therefore formulated the hypothesis that the ability of the protein surface to form such pockets may be the limiting determinant of the inherent druggability at this site.

Unlike previous standalone methods for pocket detection, we instead implemented our algorithm as a term in the Rosetta [34] energy function alongside the canonical energetic determinants of protein structure such as packing, hydrogen bonding, and solvation (see Methods section). By including this biasing term in the energy function, we may use any of the standard functionalities provided in Rosetta; inclusion of this term, meanwhile, will lead to simultaneous optimization of both "pocket score" *and* the traditional energy terms. In essence, the contribution from the "pocket" term serves as a proxy for the energy associated with binding of some (unspecified) small molecule partner.

To test the hypothesis that pocket formation may be the limiting determinant of druggability, we performed biased and unbiased simulations on the unbound conformations of Bcl-X<sub>L</sub>, targeting residues at the protein interaction site as well as at equivalent randomly selected residues elsewhere on the protein surface. Surface sites included in the random set were matched to those at the protein interaction site on the basis of their secondary structure, and further that a contacting pocket of equivalent size to that of the protein interaction site (evaluated by Q-SiteFinder [19]) was present in the unbound conformation (see Text S1 and Figure S3); further, the random sites were each at least 12 Å from one another. Both backbone and sidechain degrees of freedom were allowed to move during simulations (see Methods section). The deep pocket volumes from each of 1,000 conformations generated via each method are shown as cumulative histograms in Figure 2A. Pockets at the protein interaction site (solid red lines) form more often and are significantly larger than those formed elsewhere on the protein surface (dashed black lines). The largest pockets in the biased simulations are sampled with much higher frequency than in the corresponding unbiased simulations (Figure 2A), demonstrating that the biasing potential drives sampling towards these conformations. These observations further hold for each of the other six additional proteins comprising our test set (Figures 2B-G), and also after inclusion of additional random sites (Figure S4) or starting from the protein-bound conformation (Figure S5).

To examine the physiological relevance of the conformations generated in biased simulations, we compared their energies to those obtained in equivalent unbiased simulations. For Bcl- $X_L$ , we used Rosetta to evaluate the (unbiased) energy for each of 1000

Protein	Partner	Number of residues	Number of residues used in simulations	Deep pocket volume from unbound protein structure (Å <sup>3</sup> )	Deep pocket volume from protein-protein complex (Å <sup>3</sup> )	Deep pocket volume from inhibitor-bound complex $({\rm \AA}^3)$	Deep pocket volume from conformations generated via biased simulations (the 5% with lowest energy) $(Å^3)$	Affinity of most potent known inhibitor (nM)
MDM2	p53	492	119	129	150	201	194	80
ZipA	FtsZ	328	139	91	166	61	92	12,000
Bcl-xL	Bak	233	141	200	377	342	210	0.5
XIAP	SMAC	497	117	158	27	40	323	67
IL-2	IL-2R	153	128	38	38	38	177	60
HPV E2	HPV E1	367	193	93	92	132	129	40
FKBP12	Tbeta1-R	108	107	90	124	145	116	7



Figure 2. Surface pockets emerge only at druggable sites. Volumes of surface pockets are shown from conformations generated with no biasing potential (*left*) and upon inclusion of a "pocket

opening" biasing potential (*right*) for each of the seven proteins that comprise our test set. Surface pockets occur at druggable protein interaction sites (*solid red lines*) more frequently than elsewhere on the protein surface (*dashed black lines*). (**A**) Bcl-X<sub>L</sub>. (**B**) IL-2. (**C**) FKBP12. (**D**) HPV E2. (**E**) ZipA. (**F**) MDM2. (**G**) BIR3 domain of XIAP. doi:10.1371/journal.pcbi.1002951.g002

conformations generated from an unbiased simulation, a simulation with the biasing term applied to either the protein interaction site or a random surface residue, and equivalent simulations in which the weight of the biasing term was increased tenfold. In all cases we evaluated energies without contribution from the biasing term; a histogram of these energies is shown in Figure 3A. Conformations from the unbiased simulation (green solid line) have a very similar distribution of energies as conformations from a simulation in which the biasing potential was applied to a randomly selected target residue (black dashed line); this is unsurprising given that few of these conformations contain pockets, implying that a similar ensemble of conformations are sampled. Applying a stronger weight to the biasing potential using the same randomly selected target residue (solid black line) leads to conformations that are far less energetically favorable, indicating that pocket opening to satisfy the biasing potential could not be achieved without extensive energetic cost to the protein. In contrast, applying the biasing potential to a residue at the protein interaction site led to conformations that were only slightly higher in energy (and with overlapping distributions) than those conformations sampled in the unbiased simulation, for either weight of the biasing potential (red lines). A scatterplot showing the deep pocket volume for each of these conformations highlights the fact that conformations containing large pockets are sampled with the moderate biasing potential only if it is applied at the protein interaction site (Figure 3B, cyan vs. red points). Application of the stronger biasing potential to random (carefully matched) surface sites leads to generations of low-energy conformations without large pockets, and also pocket-containing conformations with much higher energy (Figure 3B, blue points).

The same observations also hold for each of the other six proteins comprising our test set (**Figure S6**, **Table 1**). Collectively, these results demonstrate that pocket opening at the druggable site can occur with little energetic cost to the protein, while pocket opening elsewhere on the protein surface requires that the protein adopt a highly unfavorable conformation.

It is notable that in each of these seven examples surface pockets were identified at the protein interaction site. It is equally notable, however, that similar surface pockets were *not* observed elsewhere on the protein surface (**Figure 2**). This comparison highlights the qualitative difference between the protein interaction site – already demonstrated to be druggable in a practical sense for each of these examples – and the remainder of the protein surface, at which high-affinity interactions with small molecules have not been observed.

# Pocket shapes are encoded on the protein surface

The results presented above demonstrate that there is a natural predisposition towards pocket formation in certain druggable regions on the protein surface. We next asked whether these preferred pocket-containing conformations dictate the range of potential small-molecule inhibitors suitable for binding at this site. Should this be the case, protein conformations generated using a biasing potential to induce pocket formation should resemble inhibitor-bound conformations more than conformations generated using an equivalent unbiased protocol.

Turning first to  $Bcl-X_L$ , we aligned the lowest-energy conformation produced from the biased simulations described earlier



Energy (Rosetta energy units)

Figure 3. Energetic analysis of Bcl-X<sub>L</sub> pocket opening. (A) Conformations generated without the use of a biasing potential (solid green line) show a similar distribution of energies to those generated with the biasing potential at a randomly selected target residue (dashed black line); increasing the strength of the biasing potential here leads to conformations with higher energies (solid black line). In contrast, application of the biasing potential at the protein interaction site (red lines) leads to conformations with a distribution of energies that strongly overlaps with those energies of conformations sampled in the unbiased simulations, suggesting that these conformations represent low-energy states accessible to the unbound protein. (B) A scatterplot showing the deep pocket volume for conformations generated with the biasing potential applied to one of the random sites (moderate bias in cyan, strong bias in blue) or to the protein interaction site (moderate bias in red, strong bias in orange). Low-energy conformations containing large pockets are sampled only if the biasing potential it is applied at the protein interaction site; while large pockets are sampled using the strong bias at random sites, these conformations have considerably higher energy. All energies shown here were evaluated in the absence of the biasing potential, for fair comparison. doi:10.1371/journal.pcbi.1002951.g003

(which were started from the unbound protein structure) to both unbound and inhibitor-bound crystal structures (**Figure 4A**). Direct superposition of the inhibitor from the bound structure onto the unbound crystal structure reveals extensive steric clashes (**Figure 4B**), highlighting the local protein conformational changes that must take place in order for inhibitor binding to occur. Remarkably, examination of the conformation from the biased simulation shows that part of the protein surface has adopted a shape highly complementary to the inhibitor (**Figure 4C, right side**) – even though no information about the identity of the inhibitor was included in the simulation that produced this conformation. Further, this conformational change occurred without direct involvement of the "target" residue at which the biasing term was applied.

On the other hand, the conformation from the biased simulation did not recapitulate the bound conformation exactly:



**Figure 4. Representative conformations of Bcl-X**<sub>L</sub>. An unbound crystal structure (*pink*), an inhibitor-bound crystal structure (*green, with inhibitor shown in sticks*), and a low-energy conformation generated from the unbound crystal structure using the biasing potential (*cyan, with target residue in red*) are shown. (**A**) The overall protein architecture is preserved amongst all three; movement of the helix in the foreground upon binding is not recapitulated in the pocket-opened conformation. (**B**–**D**) The pocket revealed in this low-energy conformation nonetheless strongly resembles the surface pocket in the bound crystal structure, and even bears shape-complementarity to the inhibitor. The identity of the inhibitor was not used in generating this conformation, but was added retrospectively for visual comparison. doi:10.1371/journal.pcbi.1002951.g004

the latter contains an extension to the surface pocket (left side in the orientation shown) to accommodate the "tail" of the inhibitor (**Figure 4D**). Comparison of the protein backbone in all three structures (**Figure 4A**) reveals the protein structural reorganization required to accommodate this "tail": the helix below the inhibitor (**Figure 4A**, **foreground**) unwinds on the left side, and tightens on the right side (in the orientation shown). Because the deep pocket volume is larger in the inhibitor-bound crystal structure than in the conformation from the biased simulation, we anticipate that sampling in these simulations was insufficient to fully recapitulate this highly concerted conformational change.

The diversity of conformational changes associated with inhibitor binding in different proteins is highlighted in Figure S7. In IL-2, slight rearrangement of a helix (at top of the orientation shown) allows a sidechain rotamer change that both reveals the binding pocket and creates new interactions with inhibitor shown (Figure S7A). In contrast FKBP12, HPV E2 and ZipA each contain a pre-formed pocket on the unbound protein surface that strongly complements the inhibitor, indicating that inhibitor binding occurs primarily via a "lock-and-key" mechanism in these cases (Figure S7B-D). Expansion of the ZipA pocket observed in the biased simulations (Figure 2E) resulted from sidechain rearrangement on the surface to the left side of the binding site (in the orientation shown), without disrupting the preordered portions of the binding site. Nutlin binding to MDM2, on the other hand, requires splaying apart of two surface helices to create the binding pocket: this conformational change is dramatically recapitulated in the lowest-energy individual conformation from the biased simulations (Figure S7E). Finally, unlike these previous examples showing modest conformational changes, binding of a Smac-mimetic to the BIR3 domain of XIAP is



**Figure 5. Opening pockets shifts the conformational ensemble towards inhibitor-bound structures.** Distributions of RMSD over interface atoms (iRMSD) to the closest inhibitor-bound crystal structure for conformations generated with (*red lines*) or without (*black lines*) the biasing potential. The iRMSDs from the unbound structure to the most and least similar inhibitor-bound crystal structures are also indicated (*dashed brown vertical lines*). For all seven proteins comprising the test set, the biased simulations produced conformations closer to an inhibitor-bound crystal structure than the unbiased simulations. doi:10.1371/journal.pcbi.1002951.g005

associated with extensive rearrangement of long surface loops (Figure S7F).

To assess whether the protein conformations generated using the biasing potential resemble inhibitor-bound conformations more closely than conformations generated using an equivalent unbiased protocol, we quantitatively compared the ensembles of conformations produced by each method. We note that neither method uses any knowledge of any particular inhibitor when generating an ensemble of conformations. Therefore, rather than report the protein RMSD in reference to one pre-selected bound conformation we instead individually computed the interface RMSD (iRMSD) relative to every available inhibitor-bound crystal structure (see **Text S1**), and took the lowest of this set of iRMSD values to reflect the suitability of this conformation for binding some (known) unspecified inhibitor.

For each of the seven proteins in our test set described earlier, we show the iRMSD for each member of the biased and unbiased ensembles to its closest inhibitor-bound crystal structure (**Figure 5**). We note that the iRMSD values of conformations sampled vary across these seven different proteins; this stems from the fact discussed earlier that in some cases the unbound starting structure strongly resembles the inhibitor-bound structure (FKBP12, HPV E2, ZipA), while in other cases a more extensive conformational change accompanies binding (MDM2, XIAP) (**Figure S7**). For reference, we also indicate the iRMSD of the unbound starting structure relative to the closest and most distant inhibitor-bound structures (**Figure 5**, *dashed brown lines*) and the protein-bound structures (**Figure 5**, *dashed green lines*).

Though there is overlap between the resulting distributions, for every one of the seven proteins we examined the conformations sampled in the biased simulations are closer to an inhibitor-bound conformation than the conformations sampled in the corresponding unbiased simulations. This observation applies not just to the iRMSD of atoms directly involved in binding (Figure 5), but to the backbone atoms of the corresponding residues as well (Figure S8). Further, in five of the seven cases (Bcl-X<sub>L</sub>, IL-2, FKBP12, ZipA, and XIAP) the conformations generated from the biased simulations are closer than the starting unbound conformation to an inhibitor-bound structure (Figure 5). Conformations sampled in simulations of HPV E2 and MDM2 that moved further than the unbound conformation from the inhibitor-bound structures may be due to slight inaccuracies in the Rosetta energy function, or alternatively may be sampling novel truly "druggable" conformations for which complementary inhibitors have not yet been identified.

To further characterize the pockets generated using the biasing potential, we evaluated the percentage of solvent accessible surface area that is hydrophobic (hSASA) for each pocket (**Figure S9**). While the hydrophobicity of the inhibitor-bound pockets vary amongst the different protein test cases, in each case the pockets generated using the biasing potential exhibit similar hydrophobicity to the corresponding inhibitor-bound conformation. Like the shape of these pockets, then, their hydrophobicity appears to be an intrinsic property resulting from details of the surface geometry and composition.

Collectively these results demonstrate that biasing simulations towards conformations in which a surface pocket is present drives the resulting ensemble towards the conformations observed in inhibitor-bound crystal structures. Because no information about the identity of any particular inhibitory compound was included in the biasing potential, this suggests that the general shape (and hydrophobicity) of surface pockets available to a potential inhibitor is an inherent property of the druggable interface itself.

## Druggability of survivin

The examples comprising our test set in the studies above were selected on the basis of a known compound directly inhibiting a protein interaction. For each of these cases, then, the druggable site is coincident with the protein functional site. We next turned to survivin, an example of a protein in which the two sites are nonoverlapping.

Survivin is among the most strongly tumor-specific proteins known [35], is a notable signature of unfavorable disease outcome [36], and has been well-validated as a therapeutic target using an antisense oligonucleotide [37] and a transcriptional repressor [38,39]. Despite this intense interest, however, no direct inhibitors of survivin function have been identified. Survivin carries dual functions which together explain its important role in cancer: it serves as an inhibitor of apoptosis and is also required for cell division [40]. The anti-apoptotic activity of survivin derives from binding the Smac/DIABLO peptide via a BIR domain. While small-molecule inhibitors of other Smac-binding BIR domains have been identified [41–44] (we included the XIAP BIR domain in the test set described in previous sections), efforts in this vein using survivin have not yet proven fruitful [40,45]. Consistent with a hypothesis that the peptide-binding surface of survivin may be intrinsically undruggable, an SAR by NMR approach used by Hajduk et al. found only a single probe that interacted with this site (of 9,370 probes tested) - a full order of magnitude fewer than the number of interacting probes found using proteins for which potent inhibitors have been identified [14]. Intriguingly, this study revealed a separate distal site (Figure 6A) at which 33 probe



**Figure 6. Distinguishing druggable from functional sites on survivin.** (**A**) The crystal structure of survivin, showing the protein interaction site (*red*) and the distal druggable site identified by NMR (*blue*) [14]. (**B**) Volumes of surface pockets are compared for conformations generated with the biasing potential applied at random surface residues (*dashed black lines*), applied at the protein interaction site (*dashed red lines*), and applied at the distal druggable site (*solid blue lines*). Pockets emerge at the druggable site but not elsewhere on the protein surface.

doi:10.1371/journal.pcbi.1002951.g006

compounds bound, well within the frequency observed for druggable sites on other proteins but not useful as a starting point for modulating survivin function.

We carried out a series of biased simulations as described earlier, using the unbound crystal structure of survivin as a starting point. In agreement with results from SAR by NMR [14], we find that pockets at the "undruggable" peptide-binding site of the BIR domain do not form with any higher propensity than at randomly selected equivalent surface sites (**Figure 6B**). This result is particularly notable as it stands in contrast to our observations from the XIAP BIR domain, in which surface pockets *are* formed at the peptide-binding site (**Figure 2G**). Though beyond the scope of the current work, we anticipate that further detailed studies may reveal the precise origin of the differing propensity for binding small-molecules exhibited by the peptide-binding groove of these two highly similar BIR domains.

Though large surface pockets do not form at the peptidebinding site of survivin, they *do* form readily at a specific alternate site: the same distal surface identified by Hajduk et al. [14] (**Figure 6B**, *blue lines*). Taken together, these studies of survivin suggest that low-energy fluctuations that induce pocket formation on the protein surface are indeed a primary determinant of druggability at protein interaction sites rather than a signature of the evolved protein-binding function of these sites.

# Discussion

The results presented above support the hypothesis that the conformational transitions that open pockets at a druggable site occur with little energetic cost to the protein. This has two important consequences.

First, pocket formation within the ensemble of physiological protein fluctuations occurs only at a limited subset of surface sites. Because pocket-containing conformations are rare, it is primarily the lack of a suitable surface pocket – and not chemical composition – that renders most protein surface sites undruggable. The highly restricted druggability of protein surfaces supported by both NMR [14] and crystallographic [15] observations that probe molecules interact with a very limited subset of the protein surface. Sites at which the unbound protein crystal structure includes a preformed pocket (HPV E2, FKPB12, ZipA) typically exhibit only minor conformational changes that change the size or shape of this

pocket, and are driven primarily by sidechain reorganization. In contrast, for the examples in which the unbound protein crystal structure does not have a pre-formed pocket or has only a small surface pocket (XIAP, MDM2, Bcl-X<sub>L</sub>, IL-2) concerted motions are needed that couple backbone conformational changes to the sidechain reorganization that reveals the binding pocket. In both cases, the biasing potential we describe here serves as a proxy for the binding energy associated with some (unspecified) complementary compound, and thus drives sampling towards these conformations.

Second, these results suggest that a small number of lowresolution, low energy inhibitor shapes are encoded on the protein surface through intrinsic structural and dynamic features of the protein. We propose that low-energy fluctuations produce protein conformations displaying one of these preferred shapes, at which point a complementary small-molecule ligand may be accommodated. The protein then responds to the particular steric and chemical details of the ligand via subsequent smaller conformational changes, without changing the gross features of the pocket. This model implies that exploring the pocket shapes may give clues at low-resolution to the shapes of complementary ligands, and may form the basis for a computational screening approach that matches pocket shapes to potential inhibitory ligands.

As pointed out by Cheng and colleagues [46], pocket formation is necessary but may not be sufficient for a protein surface to be druggable: the curvature and hydrophobicity of the pocket are also important. By applying their methodology to the pocket conformations generated using the biasing potential described here, it may be possible to filter for the most druggable conformations from among these pocket-containing ensembles. These conformations may not necessarily correspond to the protein-bound conformation, which has been used in some cases as a starting point for mimicry by small-molecules [47]. Generating an ensemble of druggable conformations for a given protein target may prove valuable in identifying new inhibitory compounds, even in cases where one or more inhibitors are already known. Rather than aiming to identify additional inhibitors by analogy to these known compounds, or even using the protein structure solved in complex with one of these known compounds (or the natural protein partner), matching directly against an ensemble of pocketcontaining conformations removes bias towards this parent compound (or protein partner). We anticipate that the surface pockets to which the protein is most predisposed, as revealed by this approach and therefore identified without artifacts of conformational changes in response to any particular binding partner, may serve as an optimal starting point for computational screening and may facilitate identification of chemotypes unrelated to those of known inhibitors ("scaffold hopping" [3,48]). These new scaffolds may in turn yield potent novel small molecule inhibitors through subsequent chemical elaboration.

In principle, fragment-based approaches allow multiple probe molecules to be subsequently linked together to create a single compound presenting these probe moieties in the appropriate orientation [49]. Computational solvent mapping [27], while suitable for assessing druggability in certain cases, may prove limiting for early drug discovery because of its inability to explore the concerted backbone motions necessary for thoroughly sampling the "pocket ensemble". While FKBP sidechain motions within 6 Å of the binding site are sufficient for recognizing small probe molecules [49], the further necessity of backbone reorganization in other cases described here (XIAP, MDM2, Bcl-X<sub>L</sub>, IL-2) underscores the importance of backbone motions for conformational fluctuations that reveal surface pockets. Exploring these alternate conformations may further prove useful in guiding efforts to improve potency of known inhibitors, by identifying sites at which substituents can make additional strong contacts with the protein surface.

Finally, we note that the generality of our approach is conducive to its large-scale application for comparing druggability across many protein interaction sites. In the future, we expect this approach may be used to address the outstanding question of whether these few successful examples of small-molecules disrupting protein interaction sites are outliers, or whether they instead represent the first step towards an important class of new tools for therapeutic intervention.

#### Methods

## Identifying pockets on protein surfaces

We implemented our modified version of the LIGSITE<sup>cs</sup> [28] algorithm in the Rosetta software suite [34]. Briefly, a grid is centered at the residue of interest on the protein surface and grid points are marked as occupied by either protein (P) or solvent (S). The algorithm performs linear searches in the X, Y and Z directions as well as in each diagonal for "P-nS-P" events: cases where a line no longer than 12 Å can be drawn between two points on the protein surface that pass through only solvent, establishing this solvent region as part of a surface pocket. Subsequent additional criteria were used to eliminate spurious definition of pockets and reduce the effect of grid-based artifacts (described in **Text S1**). Adjoining grid points defined as "pocket" were clustered to determine the deep pocket volume of the largest contiguous single pocket in contact with the target residue.

#### Simulation protocol

Simulations were carried out using the "relax" protocol [50] in Rosetta, which incorporates both backbone and sidechain degrees of freedom in a Monte Carlo search. To bias the simulations towards pocket-containing conformations we added to the standard energy function an additional energy term corresponding to the current deep pocket volume multiplied by a proportionality constant of -0.25 Rosetta energy units per Å<sup>3</sup> ("moderate" bias). Simulations carried out with the "strong" bias used a proportionality constant of -0.25 Rosetta energy units per Å<sup>3</sup>. Data for each histogram was collected from 1,000 independent simulations. The simulation protocol is described in complete detail in **Text S1**.

# **Supporting Information**

Figure S1 Identifying pockets on protein surfaces (complements Figure 1). A grid is centered at the residue of interest on the protein surface (only partial grid is shown here). Grid points are classified as either "protein" (grey background) or "solvent" (white background). Linear searches (red lines) are used to identify and mark "protein-surface-protein" events (green gridpoints). These are further classified based on degree of buried (light vs. dark green) and adjoining grid points are clustered to yield discrete "pockets". Only a surface pocket in contact with the surface of the "target" residue (orange) contributes to the biasing potential. (EPS)

**Figure S2 Deep pocket volumes compared to Q-Site-Finder pocket volumes (complements Figure 1).** Deep pocket volumes of surface pockets at protein interaction sites harboring a bound inhibitor (*red circles*) and pockets found elsewhere on the protein surface (*black x's*) are plotted against the corresponding pocket volumes identified by Q-SiteFinder, for each of the sites used in **Figure 1B**. While the two are correlated, Q- SiteFinder volumes are typically larger than the corresponding deep pocket volumes. (EPS)

Figure S3 Matching of random surface sites to protein interaction sites (complements Figure 2). The distribution of pocket volumes identified by Q-SiteFinder for the complete set of random surface sites matched to the protein interaction sites on the basis of burial and secondary structure is shown (*black*); the corresponding pocket volumes for the protein interaction sites are also shown (*red arrows*). In order to create a matched set, only random sites with Q-SiteFinder pocket volumes greater than 100 Å<sup>3</sup> were considered for **Figures 2** and **3**. (EPS)

Figure S4 Surface pockets emerge only at druggable sites, upon inclusion of additional random surface sites (complements Figure 2). Pocket opening is not observed at random surface sites upon inclusion of additional sites that were previously excluded on the basis of Q-SiteFinder pocket volumes less than 100 Å<sup>3</sup> (*dashed black lines*). Symbols are as defined in Figure 2. (A) Bcl-X<sub>L</sub>. (B) IL-2. (C) FKBP12. (D) HPV E2. (E) ZipA. (F) MDM2. (G) BIR3 domain of XIAP. (EPS)

Figure S5 Surface pockets emerge at druggable sites, when starting from protein-bound conformations (complements Figure 2). Simulations were carried out only applying the biasing potential to the protein interaction site (*red*). With the exception of XIAP, results are in agreement with simulations started from the unbound protein structures (Figure 2). (A) Bcl-X<sub>L</sub>. (B) IL-2. (C) FKBP12. (D) HPV E2. (E) ZipA. (F) MDM2. (G) BIR3 domain of XIAP. (EPS)

Figure S6 Energetic analysis of pocket opening for the other members of our test set (complements Figure 3). (A) Conformations generated using the biasing potential typically have a distribution of energies that overlaps those generated with the biasing potential, suggesting that these conformations represent low-energy states accessible to the unbound protein. (B) As with Bcl-xL, low-energy conformations containing large pockets are not observed for the other members of our test set unless the biasing potential is applied to the protein interaction site. Symbols are as defined in Figure 3. (TIF)

Figure S7 Representative conformations generated using the biasing potential (complements Figure 4). An unbound crystal structure (*pink*), an inhibitor-bound crystal structure (green, with inhibitor shown in sticks), and a low-energy conformation generated from the unbound crystal structure using the biasing potential (*cyan*, with target residue in red) are shown for each of the proteins comprising our test set (except Bcl-X<sub>L</sub>, shown in Figure 4). (A) IL-2. (B) FKBP12. (C) HPV E2. (D) ZipA. (E) MDM2. (F) BIR3 domain of XIAP. (TIF)

Figure S8 Opening pockets shifts the protein backbone towards inhibitor-bound structures (complements Figure 5). Distributions of backbone RMSD over interface residues to the closest inhibitor-bound crystal structure for conformations generated with (*red lines*) or without (*black lines*) the biasing potential. The backbone RMSD over interface residues from the unbound structure to the most and least similar inhibitorbound crystal structures are also indicated (*dashed brown vertical lines*). (EPS) Figure S9 Hydrophobicity of pockets generated using the biasing potential are similar to the corresponding inhibitor-bound pockets (complements Figure 5). Distributions of percentage hydrophobic solvent accessible surface area (%hSASA) over interface atoms for conformations generated with the biasing potential (*red lines*). The lowest and highest %hSASAs from amongst all available inhibitor-bound crystal structures are indicated (*dashed green vertical lines*). (A) Bcl-X<sub>L</sub>. (B) IL-2. (C) FKBP12. (D) HPV E2. (E) ZipA. (F) MDM2. (G) BIR3 domain of XIAP.

(EPS)

**Text S1 Supplementary methods.** This supporting text contains a complete description of methodology used, including

#### References

- Sun H, Stuckey JA, Nikolovska-Coleska Z, Qin D, Meagher JL, et al. (2008) Structure-based design, synthesis, evaluation, and crystallographic studies of conformationally constrained Smac mimetics as inhibitors of the X-linked inhibitor of apoptosis protein (XIAP). J Med Chem 51: 7169–7180.
- Bruncko M, Oost TK, Belli BA, Ding H, Joseph MK, et al. (2007) Studies leading to potent, dual inhibitors of Bcl-2 and Bcl-xL. J Med Chem 50: 641– 662.
- Rush TS, 3rd, Grant JA, Mosyak L, Nicholls A (2005) A shape-based 3-D scaffold hopping method and its application to a bacterial protein-protein interaction. Journal of medicinal chemistry 48: 1489–1495.
- Wang Y, Coulombe R, Cameron DR, Thauvette L, Massariol MJ, et al. (2004) Crystal structure of the E2 transactivation domain of human papillomavirus type 11 bound to a protein interaction inhibitor. J Biol Chem 279: 6976–6985.
- Thanos CD, Randal M, Wells JA (2003) Potent small-molecule binding to a dynamic hot spot on IL-2. J Am Chem Soc 125: 15280–15281.
- Becker JW, Rotonda J, Cryan JG, Martin M, Parsons WH, et al. (1999) 32-Indolyl ether derivatives of ascomycin: three-dimensional structures of complexes with FK506-binding protein. J Med Chem 42: 2798–2804.
- Vassilev LT, Vu BT, Graves B, Carvajal D, Podlaski F, et al. (2004) In vivo activation of the p53 pathway by small-molecule antagonists of MDM2. Science 303: 844–848.
- Wells JA, McClendon CL (2007) Reaching for high-hanging fruit in drug discovery at protein-protein interfaces. Nature 450: 1001–1009.
- Alsmadi O, Herz R, Murphy E, Pinter A, Tilley SA (1997) A novel antibodydependent cellular cytotoxicity epitope in gp120 is identified by two monoclonal antibodies isolated from a long-term survivor of human immunodeficiency virus type 1 infection. J Virol 71: 925–933.
- Domling A (2008) Small molecular weight protein-protein interaction antagonists: an insurmountable challenge? Curr Opin Chem Biol 12: 281–291.
- Rechfeld F, Gruber P, Hofmann J, Kirchmair J (2011) Modulators of proteinprotein interactions: novel approaches in targeting protein kinases and other pharmaceutically relevant biomolecules. Curr Top Med Chem 11: 1305–1319.
- Zinzalla G, Thurston DE (2009) Targeting protein-protein interactions for therapeutic intervention: a challenge for the future. Future Med Chem 1: 65–93.
- Kenny CH, Ding W, Kelleher K, Benard S, Dushin EG, et al. (2003) Development of a fluorescence polarization assay to screen for inhibitors of the FtsZ/ZipA interaction. Anal Biochem 323: 224–233.
- Hajduk PJ, Huth JR, Fesik SW (2005) Druggability indices for protein targets derived from NMR-based screening data. Journal of medicinal chemistry 48: 2518–2525.
- Mattos C, Bellamacina CR, Peisach E, Pereira A, Vitkup D, et al. (2006) Multiple solvent crystal structures: probing binding sites, plasticity and hydration. J Mol Biol 357: 1471–1482.
- Metz A, Pfleger C, Kopitz H, Pfeiffer-Marek S, Baringhaus KH, et al. (2012) Hot spots and transient pockets: predicting the determinants of small-molecule binding to a protein-protein interface. J Chem Inf Model 52: 120–133.
- Seco J, Luque FJ, Barril X (2009) Binding site detection and druggability index from first principles. Journal of medicinal chemistry 52: 2363–2371.
- Eyrisch S, Helms V (2007) Transient pockets on protein surfaces involved in protein-protein interaction. J Med Chem 50: 3457–3464.
- Laurie AT, Jackson RM (2005) Q-SiteFinder: an energy-based method for the prediction of protein-ligand binding sites. Bioinformatics 21: 1908–1916.
- Foster TJ, MacKerell AD, Jr., Guvench O (2012) Balancing target flexibility and target denaturation in computational fragment-based inhibitor discovery. Journal of computational chemistry 33: 1880–1891.
- Clodfelter KH, Waxman DJ, Vajda S (2006) Computational solvent mapping reveals the importance of local conformational changes for broad substrate specificity in mammalian cytochromes P450. Biochemistry 45: 9393–9407.
- 22. Landon MR, Amaro RE, Baron R, Ngan CH, Ozonoff D, et al. (2008) Novel druggable hot spots in avian influenza neuraminidase H5N1 revealed by computational solvent mapping of a reduced and representative receptor ensemble. Chemical biology & drug design 71: 106–116.

PDB structures used in calculations. A description of pocketopened conformations for each protein in our test set is also included in this text.

(DOC)

# Acknowledgments

We thank Ragul Gowthaman and Karen Khar for valuable discussions.

# **Author Contributions**

Conceived and designed the experiments: DKJ JK. Performed the experiments: DKJ. Analyzed the data: DKJ JK. Wrote the paper: DKJ JK.

- Landon MR, Lancia DR, Jr., Yu J, Thiel SC, Vajda S (2007) Identification of hot spots within druggable binding regions by computational solvent mapping of proteins. Journal of medicinal chemistry 50: 1231–1240.
- Prasad JC, Goldstone JV, Camacho CJ, Vajda S, Stegeman JJ (2007) Ensemble modeling of substrate binding to cytochromes P450: analysis of catalytic differences between CYP1A orthologs. Biochemistry 46: 2640–2654.
- Eyrisch S, Helms V (2009) What induces pocket openings on protein surface patches involved in protein-protein interactions? J Comput Aided Mol Des 23: 73–86.
- Bakan A, Nevins N, Lakdawala AS, Bahar I (2012) Druggability Assessment of Allosteric Proteins by Dynamics Simulations in the Presence of Probe Molecules. Journal of chemical theory and computation 8: 2435–2447.
- Kozakov D, Hall DR, Chuang GY, Cencic R, Brenke R, et al. (2011) Structural conservation of druggable hot spots in protein-protein interfaces. Proc Natl Acad Sci U S A 108: 13528–13533.
- Huang B, Schroeder M (2006) LIGSITEcsc: predicting ligand binding sites using the Connolly surface and degree of conservation. BMC structural biology 6: 19.
- Brady GP, Jr., Stouten PF (2000) Fast prediction and visualization of protein binding pockets with PASS. J Comput Aided Mol Des 14: 383–401.
- Hendlich M, Rippmann F, Barnickel G (1997) LIGSITE: automatic and efficient detection of potential small molecule-binding sites in proteins. Journal of Molecular Graphics & Modelling 15: 359–363, 389.
- Liang J, Edelsbrunner H, Woodward C (1998) Anatomy of protein pockets and cavities: measurement of binding site geometry and implications for ligand design. Protein Sci 7: 1884–1897.
- Peters KP, Fauck J, Frommel C (1996) The automatic search for ligand binding sites in proteins of known three-dimensional structure using only geometric criteria. J Mol Biol 256: 201–213.
- Weisel M, Proschak E, Schneider G (2007) PocketPicker: analysis of ligand binding-sites with shape descriptors. Chem Cent J 1: 7.
- Leaver-Fay A, Tyka M, Lewis SM, Lange OF, Thompson J, et al. (2011) ROSETTA3: an object-oriented software suite for the simulation and design of macromolecules. Methods Enzymol 487: 545-574.
- Velculescu VE, Madden SL, Zhang L, Lash AE, Yu J, et al. (1999) Analysis of human transcriptomes. Nat Genet 23: 387–388.
- Paik S, Shak S, Tang G, Kim C, Baker J, et al. (2004) A multigene assay to predict recurrence of tamoxifen-treated, node-negative breast cancer. N Engl J Med 351: 2817–2826.
- Rodel F, Frey B, Leitmann W, Capalbo G, Weiss C, et al. (2008) Survivin antisense oligonucleotides effectively radiosensitize colorectal cancer cells in both tissue culture and murine xenograft models. Int J Radiat Oncol Biol Phys 71: 247–255.
- Lewis KD, Samlowski W, Ward J, Catlett J, Cranmer L, et al. (2011) A multicenter phase II evaluation of the small molecule survivin suppressor YM155 in patients with unresectable stage III or IV melanoma. Invest New Drugs 29: 161– 166.
- Kita A, Nakahara T, Yamanaka K, Nakano K, Nakata M, et al. (2011) Antitumor effects of YM155, a novel survivin suppressant, against human aggressive non-Hodgkin lymphoma. Leukemia research 35: 787–792.
- Altieri DC (2008) Survivin, cancer networks and pathway-directed drug discovery. Nature reviews Cancer 8: 61–70.
- Sun H, Lu J, Liu L, Yi H, Qiu S, et al. (2010) Nonpeptidic and potent smallmolecule inhibitors of cIAP-1/2 and XIAP proteins. J Med Chem 53: 6361– 6367.
- Dean EJ, Ward T, Pinilla C, Houghten R, Welsh K, et al. (2010) A small molecule inhibitor of XIAP induces apoptosis and synergises with vinorelbine and cisplatin in NSCLC. Br J Cancer 102: 97–103.
- Wang Z, Cuddy M, Samuel T, Welsh K, Schimmer A, et al. (2004) Cellular, biochemical, and genetic analysis of mechanism of small molecule IAP inhibitors. J Biol Chem 279: 48168–48176.
- Schimmer AD, Welsh K, Pinilla C, Wang Z, Krajewska M, et al. (2004) Smallmolecule antagonists of apoptosis suppressor XIAP exhibit broad antitumor activity. Cancer Cell 5: 25–35.

- 45. Wendt MD, Sun C, Kunzer A, Sauer D, Sarris K, et al. (2007) Discovery of a novel small molecule binding site of human survivin. Bioorg Med Chem Lett 17: 3122–3129.
- Cheng AC, Coleman RG, Smyth KT, Cao Q, Soulard P, et al. (2007) Structurebased maximal affinity model predicts small-molecule druggability. Nature Biotechnology 25: 71–75.
- Koes D, Khoury K, Huang Y, Wang W, Bista M, et al. (2012) Enabling largescale design, synthesis and validation of small molecule protein-protein antagonists. PLoS One 7: e32839.
- Quintus F, Sperandio O, Grynberg J, Petitjean M, Tuffery P (2009) Ligand scaffold hopping combining 3D maximal substructure search and molecular similarity. BMC Bioinformatics 10: 245.
- Shuker SB, Hajduk PJ, Meadows RP, Fesik SW (1996) Discovering high-affinity ligands for proteins: SAR by NMR. Science 274: 1531–1534.
- Qian B, Raman S, Das R, Bradley P, McCoy AJ, et al. (2007) High-resolution structure prediction and the crystallographic phase problem. Nature 450: 259– 264.